Author's response to reviews

Title: Higher Anti-Depressant Dose and Major Adverse Outcomes in Moderate Chronic Kidney Disease: A Retrospective Population-Based Study

Authors:

Varun Dev (vdev@uwo.ca)
Stephanie N Dixon (stephanie.dixon@ices.on.ca)
Jamie L Fleet (Jamie.Fleet@lhsc.on.ca)
Sonja Gandhi (Sonja.Gandhi@lhsc.on.ca)
Tara Gomes (GomesT@smh.ca)
Ziv Harel (HarelZ@smh.ca)
Arsh K Jain (Arsh.Jain@lhsc.on.ca)
Salimah Z Shariff (Salimah.Shariff@ices.on.ca)
Davy Tawadrous (Davy.Tawadrous@lhsc.on.ca)
Matthew A Weir (Matthew.Weir@lhsc.on.ca)
Amit X Garg (Amit.Garg@lhsc.on.ca)

Version: 2 Date: 8 January 2014

Author's response to reviews:

n/a